\documentclass[10pt, english]{amsart}

\usepackage{amsmath,amssymb,enumerate}

\usepackage[T1]{fontenc}
%\usepackage[utf8]{inputenc}
\usepackage[all]{xy}

\usepackage{babel}
\usepackage{amstext}
\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{latexsym}
\usepackage{ifthen}

\usepackage{xypic}
\xyoption{all}
\pagestyle{plain}

\newcommand{\cal}{\mathcal}


\newcommand{\id}{{\rm id}}
\renewcommand{\O}{{\mathcal O}}
\newcommand{\I}{{\mathcal I}}
\newcommand{\J}{{\mathcal J}}
\newcommand{\E}{{\mathcal E}}
\newcommand{\PN}{{\mathbb P}}
\newcommand{\AN}{{\mathbb A}}
\newcommand{\PF}{{\mathbb F}}
\newcommand{\rk}{{\rm rk}}
\newcommand{\codim}{{\rm codim}}
%\newcommand{\Pic}{{\rm Pic}}
\newcommand{\Cl}{{\rm Cl}}
\newcommand{\lra}{\longrightarrow}
\newcommand{\im}{{\rm im}}
\newcommand{\Bs}{{\rm Bs}}
\newcommand{\sJ}{{\mathcal J}}
\newcommand{\sK}{{\mathcal K}}


%\renewcommand{\theequation}{\arabic{section}.\arabic{equation}}

\newtheorem{lemma1}{}[section]


\newenvironment{lemma}{\begin{lemma1}{\bf Lemma.}}{\end{lemma1}}
\newenvironment{example}{\begin{lemma1}{\bf Example.}\rm}{\end{lemma1}}
\newenvironment{abs}{\begin{lemma1}\rm}{\end{lemma1}}
\newenvironment{theorem}{\begin{lemma1}{\bf Theorem.}}{\end{lemma1}}
\newenvironment{proposition}{\begin{lemma1}{\bf Proposition.}}{\end{lemma1}}
\newenvironment{corollary}{\begin{lemma1}{\bf Corollary.}}{\end{lemma1}}
\newenvironment{remark}{\begin{lemma1}{\bf Remark.}\rm}{\end{lemma1}}
\newenvironment{remarks}{\begin{lemma1}{\bf Remarks.}\rm}{\end{lemma1}}
\newenvironment{definition}{\begin{lemma1}{\bf Definition.}}{\end{lemma1}}
\newenvironment{notation}{\begin{lemma1}{\bf Notation.}}{\end{lemma1}}
\newenvironment{case}{\begin{lemma1}{\bf Case.}}{\end{lemma1}}
\newenvironment{setup}{\begin{lemma1}{\bf Setup.}}{\end{lemma1}}
\newenvironment{construction}{\begin{lemma1}{\bf Construction.}\rm}{\end{lemma1}}
\newenvironment{conjecture}{\begin {lemma1}{\bf Conjecture.}}{\end{lemma1}}
\newenvironment{question}{\begin{lemma1}{\bf Question.}}{\end{lemma1}}
\newenvironment{problem}{\begin{lemma1}{\bf Problem.}}{\end{lemma1}}
\newenvironment{assumption}{\begin{lemma1}{\bf Assumption.}}{\end{lemma1}}

\newenvironment{remark*}{{\bf Remark.}}{}
\newenvironment{example*}{{\bf Example.}}{}
\newenvironment{assumption*}{{\bf Assumption.}}{}


\renewcommand{\labelenumi}{\alph{enumi})}
\renewcommand{\labelenumii}{(\roman{enumii})}

\newcommand{\R}{\ensuremath{\mathbb{R}}}
\newcommand{\Q}{\ensuremath{\mathbb{Q}}}
\newcommand{\Z}{\ensuremath{\mathbb{Z}}}
\newcommand{\C}{\ensuremath{\mathbb{C}}}
\newcommand{\N}{\ensuremath{\mathbb{N}}}
\newcommand{\PP}{\ensuremath{\mathbb{P}}}
\newcommand{\D}{\ensuremath{\mathbb{D}}}
\newcommand{\G}{\ensuremath{\mathbb{G}}}

%\newcommand{\rk}{\ensuremath{rk} \ }
%\newcommand{\im}{\ensuremath{\mbox{im} \hspace{0.5ex}}}
\newcommand{\alb}[1]{ \ensuremath{ \mbox{alb}_{#1}  } }
\newcommand{\Alb}[1]{\ensuremath{\mbox{Alb}(#1)\hspace{0.5ex}}}

\newcommand{\dimension}[1]{\ensuremath{\mbox{dim}(#1)}}

%\newcommand{\NS}[1]{ \ensuremath{ \mbox{NS}(#1) } }
%\newcommand{\Pic}[1]{ \ensuremath{ \mbox{Pic}(#1) } }
\newcommand{\Osheafo}{\ensuremath{ \mathcal{O} }}
\newcommand{\Homsheaf} { \ensuremath{ \mathcal{H} \! om}}
\newcommand{\Ssheaf} { \ensuremath{ \mathcal{S} } }
\newcommand{\Isheaf} { \ensuremath{ \mathcal{I} } }
\newcommand{\Tsheaf} { \ensuremath{ \mathcal{T} } }
\newcommand{\Osheaf} { \ensuremath{ \mathcal{O} } }
\newcommand{\Esheaf} { \ensuremath{ \mathcal{E} } }
\newcommand{\Vsheaf} { \ensuremath{ \mathcal{V} } }
\newcommand{\Mor}[1]{ \ensuremath{ \mbox{Mor}(#1) } }

%\newcommand{}[]{}





\newcommand{\merom}[3]{\ensuremath{#1:#2 \dashrightarrow #3}}

\newcommand{\holom}[3]{\ensuremath{#1:#2  \rightarrow #3}}
\newcommand{\fibre}[2]{\ensuremath{#1^{-1} (#2)}}


\makeatletter
\ifnum\@ptsize=0 \addtolength{\hoffset}{-0.3cm} \fi \ifnum\@ptsize=2 \addtolength{\hoffset}{0.5cm} \fi \sloppy




%script letters
\newcommand\sA{{\mathcal A}}
\newcommand\sB{{\mathcal B}}
\newcommand\sE{{\mathcal E}}
\newcommand\sExt{{\mathcal Ext}}
\newcommand\sF{{\mathcal F}}
\newcommand\sG{{\mathcal G}}
\newcommand\sH{{\mathcal H}}
\newcommand\sI{{\mathcal I}}
\newcommand\sQ{{\mathcal Q}}
\newcommand\sJT{{\mathcal JT}}
%\newcommand\sK{{\mathcal K}}
\newcommand\sL{{\mathcal L}}
\newcommand\sN{{\mathcal N}}
\newcommand\sO{{\mathcal O}}
\newcommand\sP{{\mathcal P}}
\newcommand\sS{{\mathcal S}}
\newcommand\sV{{\mathcal V}}
\newcommand\sX{{\mathcal X}}
\newcommand\sC{{\mathcal C}}
\newcommand\bR{{\mathbb R}}
\newcommand\bZ{{\mathbb Z}}
\newcommand\bC{{\mathbb C}}
\newcommand\bQ{{\mathbb Q}}
\newcommand\bN{{\mathbb N}}
\newcommand\Hom{{\rm Hom}}
\newcommand\Ext{{\rm Ext}}
\newcommand\sR{{\mathcal R}}
\newcommand\sM{{\mathcal M}}

\newcommand\sD{{\mathcal D}}


\newcommand\sLe{{L_{\varepsilon}}}
\newcommand\sLee{{L^3_{\varepsilon}}}
\newcommand\sLef{{L^2_{\varepsilon}}}


\newcommand\RO{{\mathfrak R\mathcal O}}

\newcommand\pp{{\overline \partial}}



\DeclareMathOperator*{\Irr}{Irr}
\DeclareMathOperator*{\Sym}{Sym}
\DeclareMathOperator*{\pic}{Pic}
\DeclareMathOperator*{\sing}{sing}
\DeclareMathOperator*{\Pic0}{Pic^0}
\DeclareMathOperator*{\Nm}{Nm}
\DeclareMathOperator*{\red}{red}
\DeclareMathOperator*{\num}{num}
\DeclareMathOperator*{\norm}{norm}
\DeclareMathOperator*{\nons}{nons}
\DeclareMathOperator*{\lin}{lin}

\newcommand\Ctwo{\ensuremath{C^{(2)}}}
\newcommand{\chow}[1]{\ensuremath{\mathcal{C}(#1)}}
\newcommand{\chown}[1]{\ensuremath{\mathcal{C}_n(#1)}}
\newcommand{\douady}[1]{\ensuremath{\mathcal{D}(#1)}}
\newcommand{\Chow}[1]{\ensuremath{\mbox{\rm Chow}(#1)}}



\newcommand{\Comp}{\ensuremath{\mathcal{K}}}
\newcommand{\barComp}{\ensuremath{\bar{\mathcal{C}}}}
\newcommand{\barI}{\ensuremath{\bar{I}}}
\newcommand{\Afive}{\ensuremath{\mathcal{A}_5}}
\newcommand{\barAfive}{\ensuremath{\bar{\mathcal{A}_5}}}
\newcommand{\ChowTheta}[1]{\ensuremath{\mathcal{C}_{\frac{\Theta^3}{3!}}(#1)}}
\newcommand{\ChowTwo}{\ensuremath{\mathcal{C}_{\frac{\Theta^2}{2!}}}}
\newcommand{\ChowbarI}{\ensuremath{\mathcal{C}_{\frac{\Theta^3}{3!}}(U/\barI)}}

\DeclareMathOperator*{\supp}{Supp}


\setcounter{tocdepth}{2}

 \renewcommand\thesubsection{\thesection.\Alph{subsection}}
 

%\usepackage[notref,notcite]{showkeys}
 \setlength{\parindent}{0pt}
\setlength{\parskip}{\smallskipamount}


\newcommand{\NAY}{\overline{\mbox{NA}}(Y)}
\newcommand{\NEX}{\overline{\mbox{NE}}(X)}
\newcommand{\NAX}{\overline{\mbox{NA}}(X)}
\newcommand{\NE}[1]{ \ensuremath{ \overline { \mbox{NE} }(#1)} }
\newcommand{\NA}[1]{ \ensuremath{ \overline { \mbox{NA} }(#1)} }

\DeclareMathOperator*{\NS}{NS}
\DeclareMathOperator*{\Rat}{RatCurves^n}
\newcommand{\Hilb}{\ensuremath{\mathcal{H}}}
\newcommand{\Univ}{\ensuremath{\mathcal{U}}}



\title{The algebraic dimension of compact complex threefolds with vanishing second Betti number} 
\date{\today}

%\subjclass[2000]{32J27, 14E30, 14J30, 32J17, 32J25}
%\keywords{MMP, rational curves, Zariski decomposition, K\"ahler manifolds, abundance}

\author{Fr\'ed\'eric Campana}
\author{Jean-Pierre Demailly}
\author{Thomas Peternell}


\address{Fr\'ed\'eric Campana, Institut Elie Cartan,
Universit\'e Henri Poincar\'e,
BP 239,
F-54506. Vandoeuvre-les-Nancy C\'edex,
France\\
}
\email{frederic.campana@univ-lorraine.fr}

\address{Jean-Pierre Demailly, Universit\'ee de Grenoble-Alpes, Institut Fourier, UMR 5582 du C.N.R.S., 100 rue des Maths, 38610 Gi\`eres, France 
}
\email{ jean-pierre.demailly@univ-grenoble-alpes.fr
}

\address{Thomas Peternell, Mathematisches Institut, Universit\"at Bayreuth, 95440 Bayreuth, 
Germany}

\email{thomas.peternell@uni-bayreuth.de}



\begin{document}
\begin{thebibliography}{BCHM10} 


\bibitem[BHPV04]{BHPV04} Barth, W. P.; Hulek, K. ; Peters, C. A. M.; Van de Ven, A.: \textit{ Compact complex surfaces. Second edition}, Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge, {\bf 4}, Springer-Verlag, 2004.

\bibitem[CDP98]{CDP98} Campana,F.; Demailly, J.P.; Peternell,Th.: \textit {The algebraic dimension of compact complex threefolds with vanishing second Betti number}, Comp. Math. {\bf 112} (1998), 77-91 

\bibitem[DO99]{DO99} Dloussky,G.; Oeljklaus,K.: \textit{Vector fields and foliations on compact surfaces of class $VII_0$}, Ann. Inst. Fourier, {\bf 49} (1999), 1503-1545

\bibitem[DOT01]{DOT01} Dloussky,G.; Oeljklaus,K.; Toma,M.: \textit{Surfaces de la classe $VII_0$ admettant un champ de vecteurs}, Comment. Math. Helv. {\bf 76} (2001), 640-664 

\bibitem[En81]{En81} Enoki,I,:  \textit { Surfaces of class $VII_0$  with curves}, Tohoku Math. J. (2) {\bf 33} (1981), 453-492

\bibitem[Et15]{Et15}  Etesi, G.:  \textit{ Complex structure on the six dimensional sphere from a spontaneous symmetry breaking}, J. Math. Phys. {\bf 56} (2015), 21pp; Erratum ibid. 9pp

\bibitem[GH90]{GH90} Gellhaus,Ch.; Heinzner,P.:  \textit{ Komplexe Fl\"achen mit holomorphen Vektorfeldern},  Abh. Math. Sem. Univ. Hamburg {\bf 60} (1990), 37-46

\bibitem[Ha77]{Ha77} Hartshorne, R.: \textit{Algebraic geometry}, Graduate texts in mathematics, vol. {\bf 52}, Springer-Verlag, 1977

\bibitem[Ha95]{Ha95}  Hausen,J.: \textit{Zur Klassifikation glatter kompakter $\mathbb C^*$-Fl\"achen}, Math. Ann. {\bf 301} (1995), 763-769

\bibitem[In74]{In74} Inoue,M.:   \textit{On surfaces of type $VII_0$}, Invent. Math. {\bf 24} (1974), 269-310

\bibitem[KW88]{KW88} Kunz,E.; Waldi,R.: \textit{Der F\"uhrer einer Gorensteinvariet\"at}, J. reine u. angew. Math. {\bf 388} (1988), 106-115 

\bibitem[LSS18]{LSS18} Lehn, C.; Rollenske, S.; Schinko, C.,
\textit {The complex geometry of a hypothetical complex structure on $S^6$ },
Differential Geom. Appl. {\bf 57} (2018), 121-137
     
\bibitem[Mo82]{Mo82} Mori,S.: \textit{ Threefolds whose canonical bundles are not numerically effective}, Ann. Math. (2) {\bf 116} (1982), 133-176 

\bibitem[Ue75]{Ue75} Ueno,K.: \textit{Classification theory of algebraic varieties and compact complex spaces}, Lecture Notes in Math. {\bf 439}. Springer 1975

%\bibitem[Ue87]{Ue87} Ueno,K.: \textit{On compact analytic threefolds with non-trivial Albanese tori}, Math. Ann. {\bf 278}(1987), 41-70

\end{thebibliography}


\maketitle

%\tableofcontents

\section{Introduction}

The paper \cite{CDP98} studied compact complex threefolds $X$ such that the second Betti number $b_2(X) = 0.$ The main result is based on Lemma 1.5, which is certainly not
 true in general (but might still hold in the context of the paper). In any case, some of the statements and proofs need to be adapted which is done in this Corrigendum, with special regards 
 to potential complex structures on the $6$-sphere.

 
 \section{Statement of the results } 
 
We prove Theorem 2.1 in \cite{CDP98} in full generality in case $X$ has a meromorphic non-holomorphic map $X \dasharrow \mathbb P_1.$ In the remaining
case, $X$ has algebraic dimension $1$ and the algebraic reduction $f: X \to C$ is holomorphic.  In this case we prove that $c_3(X) \leq 0$; 
for simplicity, we will assume not only that $b_2(X) = 0$ but slightly stronger that $H^2(X,\mathbb Z) = 0$ and moreover that 
$H^1(X,\mathbb Z) = 0$, hence $C \simeq \mathbb P_1$. This suffices to treat the main application of 
complex structures on $S^6$. 

In summary, we shall prove 

\begin{theorem} \label{MT1}
Let $X$ be a $3-$dimensional compact complex manifold with $b_2(X) = 0$. Assume that there exists a non-holomorphic meromorphic non-constant map
$g: X \dasharrow \mathbb P_1$. 
Let $B$  be a holomorphic vector bundle on $X$. Then 
\begin{enumerate}
\item $H^{i}(X, B \otimes \sM) = 0 $ for $i \geq 0$ and $\sM \in {\rm Pic}^{\circ}(X) $ generic.
\item $\chi(X, B \otimes \sM) = 0 $ for all $\sM \in {\rm Pic}^{\circ}(X) $ 
\item $c_3(X) = 0$; i.e., either $b_1(X) = 0$ and $b_3(X) = 2$ or $b_1(X) = 1$ and $b_3(X) = 0$.
\end{enumerate}
\end{theorem}

Another proof has been given in \cite{LSS18}. 

Theorem \ref{MT1} takes care of all threefolds $X$ with $1 \leq a(X) \leq 2$ and $b_2(X) = 0$ except of those whose algebraic reduction $f: X \to C$ is holomorphic onto a curve $C$. 
In this case the general fiber has Kodaira dimension $\kappa (X_c) \leq 0$.  







By topological considerations and surface classification, the general fiber of $f$ is either a torus, a primary Kodaira surface or a surface of type VII; in the latter case it is actually a primary Hopf surface or an Inoue surface,
Lemma \ref{lemirrdcomp} and Lemma \ref{lem:fibers}. 
The case that the general fiber is a Kodaira surface is ruled out in Corollary \ref{cor:Kodaira}.
Then we show 




\begin{theorem} \label{MT2}
Let $X$ be a $3-$dimensional compact complex manifold with $H^1(X,\mathbb Z) = H^2(X,\mathbb Z) = 0$ and algebraic dimension $a(X) = 1$. Assume that
the algebraic reduction $f: X \to C $ is holomorphic.
Then
\begin{enumerate}
\item $H^{i}(X, T_X \otimes \sM) = 0 $ for $i \ne1$.
\item $\chi(X, T_X \otimes \sM) \leq 0 $.
\item $c_3(X) \leq  0$. 
\end{enumerate}
\end{theorem}



As a consequence we deduce 



\begin{corollary} Let $X$ be a 3-dimensional compact complex manifold homeomorphic to  $S^6$. Then $a(X) = 0$. 
\end{corollary} 

\begin{proof} Obviously, $a(X) \ne 3$, otherwise $X$ is Moishezon and therefore $b_2(X) \ne 0$. If $a(X) = 2,$ then there exists a meromorphic non-holomorphic map 
$g:X \to \mathbb P_1$. Then we conclude by Theorem \ref{MT1} that $c_3(X) = 0$. By Hopf's theorem, $c_3(X) = \chi_{\rm top}(S^6) = 2$, a contradiction. 
If $a(X) = 1 $ and the algebraic reduction $g: X \dasharrow C$ is not holomorphic, then $C \simeq \mathbb P_1$, and we conclude again by Theorem \ref{MT1}. 
If $a(X) = 1$ and  the algebraic reduction $g: X \dasharrow C$ is holomorphic, then we apply Theorem \ref{MT2} and obtain the same contradiction as before. 
\end{proof} 

We comment on the strategy to prove Theorem \ref{MT2}. 
The arguments of \cite{CDP98} show the following 

\begin{proposition} \label{prop:CDP}
Let $X$ be a 3-dimensional compact complex manifold with $b_2(X) = 0$ and algebraic dimension $a(X) = 1$. Assume that the algebraic reduction $f: X\to C$ is 
holomorphic. 
If  \begin{equation}  \label{eq0a} R^2f_*(T_X \otimes \sL) = 0\end{equation} 
for some (or general) $\sL \in {\rm Pic}(X)$, then the assertions of Theorem \ref{MT2} hold. 

\end{proposition} 

Equation (\ref{eq0a}) is equivalent to the vanishing 
$$ H^2(X_c,T_X \otimes \sL \vert X_c) = 0 $$
for all complex-analytic fibers $X_c$ (with the natural fiber structure) of $f$, equivalently, 
$$ H^0(X_c,\Omega^1_X \otimes \sL^* \vert X_c) = 0.$$ 
The key is to show that the restriction $\sL \vert X_c$  of some or the general line bundle $\sL$ to any fiber is never torsion. 
Then we compute directly on $X_c$; here the case when $X_c$ is singular, in particular non-normal, needs special care. 



For further informations on the problem of complex structures on $S^6$, we refer to \cite{Et15} and to volume 57 of the journal Differential Geometry and its Applications.




\section{Proof of Theorem \ref{MT1} }
Instead of simply pointing out the additions in the proof of \cite[Theorem 2.1]{CDP98} to be made, we
give full details for the benefit of the reader. 
Notice first that (\ref{MT1})(b) follows from  (\ref{MT1})(a) since $\chi(X,B \otimes \sM)$ does not depend on $\sM$, and 
 (\ref{MT1})(c) is a consequence of  (\ref{MT1})(b)
by applying Riemann-Roch to $B = T_X$ and $\sM = \sO_X$. So it remains to prove   (\ref{MT1})(a). By Serre duality,  (\ref{MT1})(a) needs only to be shown 
for $i = 0$ and $i = 2$. The case $i = 0$ follows from  \cite[Cor1.3]{CDP98}, since $X$ does carry effective non-zero divisors. Thus 
it remains to prove
\begin{equation} \label{EQ1}  H^2(X,B \otimes \sM) \end{equation}
for generic, equivalently one, line bundle $\sM$. 
Let 
$g: X \dasharrow \mathbb P_1$
be a non-constant  non-holomorphic meromorphic map and 
$ \sigma:\hat X \to X$
be a resolution of indeterminacies of $g$. Let 
$$  f: \hat X \to C \simeq \mathbb P_1$$ 
be the fiber space given by the  Stein factorization of the holomorphic map $\sigma \circ g.$ 
Replacing $g$ by the induced meromorphic map $X \dasharrow C$, we may assume from the beginning that $g$ has connected fibers, 
hence no Stein factorization has to be taken. 
Note that the canonical map 
$$ H^2(X,B \otimes \sM) \to H^2(\hat X, \sigma^*(B \otimes \sM)) $$
is injective (by the Leray spectral sequence). 
Hence Equation (\ref{EQ1}) follows from 
\begin{equation} \label{EQ2} H^2(\hat X, \sigma^*(B \otimes \sM)) = 0. \end{equation} 
Fix an ample divisor $A$ on $C$. Then $f^*(\sO_C(A))$ can be written as
\begin{equation} \label{EQX} f^*(\sO_C(A)) = \sigma^*(\sL) \otimes \sO_{\hat X}(-E) \end{equation} 
with a line bundle $\sL$ on $X$ and a suitable effective divisor $E$ which is supported on the exceptional set of $\sigma$ and projects onto $C$. 
To verify Equation (\ref{EQ2}) it suffices to show that 
\begin{equation} \label{EQ3} H^2(\hat X, \sigma^*(B \otimes \sM) \otimes \sO_{\hat X}(-tE)) = 0 \end{equation} 
for some effective divisor $E$ supported on the exceptional locus of $\sigma$ and some $t \geq 0$. 
In fact, consider the exact sequence
$$ H^2(\hat X, \sigma^*(B \otimes \sM) \otimes \sO_{\hat X}(-tE)) \to H^2(\hat X, \sigma^*(B \otimes \sM)) \to H^2(tE, \sigma^*(B \otimes \sM)) $$
and note that 
$$ H^2(tE, \sigma^*(B \otimes \sM)) = 0. $$
This last vanishing is seen as follows: let $Z_t$ be the complex subspace of $X$ defined by the ideal sheaf $\sigma_*(\sO_{\hat X}(-tE))$. Then 
by 
$$ R^q(\sigma_{\vert tE})_*(\sO_{tE}) =  0$$
for $q = 1,2$ and the Leray spectral sequence, 
$$ H^2(tE, \sigma^*(B \otimes \sM)) = H^2(Z_t, B \otimes \sM). $$
Now the last group vanishes since $\dim Z_t = 1$. 


By replacing $\sM$ by $\sM \otimes \sL^{t+k}$ for some positive integer $k$ and using (\ref{EQX}),   Equation (\ref{EQ3}) reads
\begin{equation} \label{EQ4} H^2(\hat X, \sigma^*(B \otimes \sM \otimes \sL^{t+k} \otimes \sO_{\hat X}(-tE))) = \end{equation} 
$$ =  H^2(\hat X, \sigma^*(B \otimes \sM \otimes \sL^k \otimes f^*(\sO_C(tA))). $$
By the Leray spectral sequence applied to $f$, Equation (\ref{EQ4}) comes down to verify 
\begin{equation} \label{EQ6} R^2f_*(\sigma^*(B \otimes \sM \otimes \sL^k)) = 0 \end{equation}
and 
\begin{equation} \label{EQ7} H^1(C, R^1f_*(\sigma^*(B \otimes \sM \otimes \sL^k)) \otimes \sO_C(tA)) =  0 \end{equation}
for suitable positive integers $k$ and $t$ and some line bundle $\sM$. 

To prove (\ref{EQ6}), let $C^* \subset C$ be the finite set of points $c \in C$ such that some component of the fiber $X_c$ does not meet $E$. In particular, 
if $X_c$ is irreducible, then $c \in C \setminus C^*$. Notice also that 
$$ R^2f_*(\sigma^*(B \otimes \sL^k)) \vert \{c\} \simeq H^2(\hat X_c,\sigma^*(B \otimes \sL^k)),$$
by the standard base change theorem. Applying Serre duality, we obtain
$$ H^2(\hat X_c,\sigma^*(B \otimes \sL^k)) \simeq H^0(\hat X_c,\sigma^*(B^* \otimes \sL^{-k}) \otimes \omega_{\hat X_c}) \simeq $$
$$ \simeq H^0(\hat X_c,\sigma^*(B^*) \otimes \sO_{\hat X_c}(-kE) \otimes \omega_{\hat X_c}). $$

We claim first that there is a number $k_0$ such that for $k \geq k_0$, 
\begin{equation} \label{eq:supp}  {\rm supp} \ R^2f_*(\sigma^*(B \otimes \sL^k)) \subset C^*.\end{equation} 
This is equivalent to saying that 
$$  H^0(\hat X_c, \sigma^*(B^* \otimes \sM^*) \otimes \sO_{\hat X_c}(-kE) \otimes \omega_{\hat X_c}) = 0$$ 
for $c \not \in C^*$. Fixing any point $c_0 \in C^*$, this number $k_0 = k_0(c)$ clearly exists; in case $X_c$ is reducible, we apply \cite[Prop.1.1]{CDP98}.
Hence the support of the direct image sheaf
$R^2f_*(\sigma^*(B \otimes \sL^{k_0}))$ is contained in a finite set $C_{k_0}$  in $C$. Since $\sigma^*(\sL) \vert X_c$ is effective, it follows that 
$C_{k} \subset C_{k_0}$. Thus, enlarging $k_0$ if necessary, (\ref{eq:supp}) is verified.

Hence we only need to consider the fibers $\hat X_c$ with $c \in C^*$. 
Let $P$ the set of line bundles $\sM$ of the form $$\sM = \sO_X(\sum m_i S_i)$$
with $m_i $ positive integers and $S_i$ fiber components of $f$ not meeting $E$ (the $S_i$ considered as surfaces in $X$). 

Since all line bundles $\sM \in P$ are trivial on $X \setminus f^{-1}(C^*)$, our previous considerations imply the existence of a number $k_0$ such that for all $k \geq k_0$ and all $\sM \in P$, 
$$ {\rm supp} \ R^2f_*(\sigma^*(B \otimes \sM \otimes \sL^k)) \subset C^*.$$ 
We are now going to construct a line bundle $\sM \in P$ such that 
$$ R^2f_*(\sigma^*(B \otimes \sM \otimes \sL^k)) = 0$$
for a suitable number $k$. 
Fix a point $c \in C^*$. Let $F_0 \subset X_c$ be the sub-divisor of $\hat X_c$ consisting of all components meeting $E$; let further $F_1 \subset \hat X_c$ be the sub-divisor consisting of all components which meet $F_1$ but not $E$. 
Continuing in this way we obtain a decomposition
$$ \hat X_c = \sum_{i=0}^rF_r $$
of sub-divisors $F_j \subset X_c$ who pairwise do not have common components and which have the property that all components of $F_j$ meet $F_{j-1}$ but do not meet $F_k$ for $k < j-1$. 
Now choose $m_r \gg 0 $ such that 
$$H^0(F_r, \sO_{\hat X}(-m_rF_{r-1}) \otimes \sigma^*(B^*) \otimes \omega_{\hat X_c} \vert F_r) = 0$$
This is possible by our construction. Next choose $m_{r-1} \geq m_r$ such that 
$$ H^0(F_{r-1}, \sO_{\hat X}(-m_{r-1}F_{r-2})  \otimes \sigma^*(B^*) \otimes \omega_{\hat X_c} \vert F_{r-1}) = 0.$$
Since ${\rm supp} (F_{r-2}) \cap {\rm supp}(F_r) = \emptyset$, we obtain
$$ H^0(F_{r-1} + F_r , \sO_{\hat X}(-m_{r-1}(F_{r-2} + F_{r-1} )  \otimes \sigma^*(B^*) \otimes \omega_{\hat X_c} \vert F_{r-1}) = 0.$$
Continuing in this way, we obtain a line bundle
$$ \sM'(c) = \sO_X(\sum_{i=2}^rm_iF_i)$$
(having in mind that the divisors $F_i, i \geq 1$ do not meet the exceptional locus of $\sigma$),
such that 
$$ H^0\big (\sum_{i=2}^r F_i, \sigma^*(B^* \otimes \sM'(c) )  \otimes \omega_{\hat X_c} \vert (\sum_{i=2}^r F_i) \big ) = 0.$$ 
Since the component $F_0$ meets $E$, it needs a special treatment. 
We observe that 
$$ \sO_{\hat X}(F_0) \simeq \sigma^* \big (\sO_X(\sigma(F_0) \big) \otimes \sO_{\hat X}(-E') $$
with some effective $\sigma$-exceptional divisor $E'$. Hence, choosing $m_1 \gg 0$ and setting 
$$ \sM(c) = \sM'(c) \otimes \sO_X(m_1\sigma(F_0)),$$
then 
$$ H^0 \big (\sum_{i=1}^r F_i, \sigma^*(B^* \otimes \sM(c)^*) \otimes \omega_{\hat X_c} \vert (\sum_{i=1}^r F_i) \big ) = 0.$$ 
Finally, enlarging $k$, we get 
$$ H^0(\hat X_c, \sigma^*(B^* \otimes \sM(c)^*) \otimes \sO_{\hat X}(-kE)\otimes  \omega_{\hat X_c} \vert X_c) = 0.$$ 
Setting
$$ \sM = \bigotimes_{c \in C^*} \sM(c),$$
this settles (\ref{EQ6}). \\
As to (\ref{EQ7}),  we fix $k$ as in (\ref{EQ6}) and then apply Serre's vanishing theorem to  the ample divisor $A$ to obtain $t$. 





%Now we argue by contradiction and assume that for all $k \geq k_0$ and all $\sM \in P,
%$$ {\rm supp}(R^2f_*(\sigma^*(B \otimes \sM \otimes \sL^k))) \ne 0.$$ 
%Hence there exists a point $c \in C^*$ and a number $k_1$ such that for all $k \geq k_1$ and all $\sM \in P$, we have
% $$ H^0(X_c,\sigma^*(B^* \otimes \sM^* \otimes \sL{-k}) \otimes \omega_{X_c}) \ne 0.$$
%Notice that $\sL$ is trivial on all fiber components of $f$ not meeting $E$. 
%Therefore we find a component $F_0$ of $X_c$ - with induced complex structure of $X_c$, such that 
 %$$ H^0(F_0,\sigma^*(B^* \otimes \sM^* \otimes \sL{-k}) \otimes \omega_{X_c} \vert. F_0$) \ne 0.$$
 %for all $\sM$ and infinitely many $k$ (possibly depending on $\sM$). 
 %Write $$X_c = F_0 + R,$$
% with the remainder $R$, having no joint component with $F_0$. 
%Now we take  $\sM = \sO_X(mR)$  for some large $m.$ Then $\sM_{ \vert F_0} = \sO_{F_0}(nC)$ with some effective curve $C \subset F_0$ and some positive integer $n$. 
%Thus 
%$$ H^0(F_0, B^* \otimes K_F \otimes \sO_{F_0}(-mC))  \ne 0.$$







%Thus for all $\sM$ and for all positive integers $k$, 
%$$  R^2f_*(\sigma^*(B \otimes \sM \otimes \sL^k)) \ne 0.$$
%Consider an irreducible fiber $\hat X_c$  of $f$. Since $E \cap \hat X_c \ne \emptyset$, 
%$$ H^2(\hat X_c,  \sigma^*(B \otimes \sM \otimes \sL^k)) = H^0(\hat X_c, \sigma^*(B^* \otimes \sM^*) \otimes \sO_{\hat X_c}(-kE) \otimes \omega_{\hat X_c}) = 0$$
%for $k \geq k_0(\sM)$. 
%By semi-continuity, there exists a number $k_0(\sM)$ such that for $k \geq k_0$, the sheaf
%$$   R^2\sigma_*(B \otimes \sM \otimes \sL^k) $$
%is supported on the finite subset of $C$ consisting of those $c$ such that the fiber of $c$ is reducible.
%Actually, applying \cite[Prop.1.1]{CDP98},  the sheaf $R^2\sigma_*(B \otimes \sM \otimes \sL^k) $ is supported on the subset of those $c$ such that 
%the fiber over $c$ contains an irreducible component not meeting $E$. 
%Since we argue by contradiction, we find for any $\sM$ a complex-analytic reducible fiber $F$ of $f$ such that 
%$$ H^2(F, \sigma^*(B \otimes \sM \otimes \sL^k)) = H^0(F,  \sigma^*(B^* \otimes \sM^* \otimes \sL^{-k}) \otimes \omega_F)) \ne 0$$
%and infinitely many $k$. Since there are only finitely many reducible fibers, $F$ can be chosen independently of $\sM$ and also the infinite set
%does not depend on $F$. 
%Moreover, $F$ has the form 
%$$ F = F_0 + R$$
%where $R$ is a divisor without joint component with $F$ and 
%$$ H^0(F_0, \sigma^*(B^* \otimes \sM^*  \otimes \sL^{-k}) \otimes \omega_F \vert F_0) \ne 0 $$
%for infinitely many $k$. By semicontinuity this then holds for al $\sM$. 
%Now we take  $\sM = \sO_X(mR)$  for some large $m.$ Then $\sM_{ \vert F_0} = \sO_{F_0}(nC)$ with some effective curve $C \subset F_0$ and some positive integer $n$. 
%Thus 
%$$ H^0(F_0, B^* \otimes K_F \otimes \sO_{F_0}(-mC))  \ne 0.$$
%But this is not possible if $m \gg 0.$ 


\section{General structure of the fibers}

From now on - for the rest of the paper -  we let $X$ be a compact complex manifold of dimension $3$ with 
$$H^1(X,\mathbb Z) = H^2(X,\mathbb Z) = 0$$ and holomorphic algebraic reduction $f:X \to C$ to the 
curve $C \simeq \mathbb P_1$.  We will freely use the theory of compact complex surfaces, in particular of non-K\"ahler surfaces, and refer to 
\cite{BHPV04} as general reference. 

An application of Riemann-Roch gives 

\begin{lemma} $\chi(D,\sO_D) = 0$ for all effective divisors $D$ on $X$. 
\end{lemma} 

\begin{proof} By Riemann-Roch,
$$ \chi(X,\sO_X(-D)) = \chi(X,\sO_X),$$
hence 
$$\chi(D,\sO_D) = \chi(X,\sO_X) - \chi(X,\sO_X(-D)) = 0.$$
\end{proof} 



\begin{lemma} \label{lemirrdcomp} 
Let $s$ be the number of singular fibers and $r$ be the numbers of irreducible components of the singular fibers. Then 
$$ r = s - 1 + b_1(X_c),$$
where $X_c$ is a smooth fiber. 
Moreover $H_1(X_c, \mathbb Z) $ is torsion free for all smooth fibers $X_c$. 
\end{lemma} 

\begin{proof} The first assertion is \cite[Lemma 3.2]{CDP98}. For the second, fix a smooth fiber $X_c$ and let $A \subset $ the union of all singular fibers of $f$ and set $X' = X \setminus A$. 
As seen in the proof of  \cite[Lemma 3.2]{CDP98}, 
$$ H_1(X',\mathbb Z) \simeq H_1(C \setminus f(A),\mathbb Z) \oplus H_1(X_c,\mathbb Z),$$
hence it suffices to show that $ H_1(X',\mathbb Z)$ is torsion free. 
To do this,  we consider the cohomology sequence for pairs, 
$$ 0 = H^4(X,\mathbb Z) \to H^4(A,\mathbb Z) \to H^5(X,A,\mathbb Z) \to H^5(X,\mathbb Z) \to 0.$$
Notice first that  $H^4(A,\mathbb Z)$ is torsion free. Further,
$H^5(X,\mathbb Z)$
is torsion free by the universal coefficent theorem, since $H_4(X,\mathbb Z)$ is torsion free: by Poincar\'e duality,
$$ H_4(X,\mathbb Z) \simeq H^2(X,\mathbb Z) = 0.$$  
Actually, $H^5(X,\mathbb Z) = 0$. 
Consequently, $$H^5(X,A,\mathbb Z) \simeq H_1(X',\mathbb Z)$$ is torsion free.
\end{proof} 
 


\begin{lemma} \label{lem:fibers} 
Let $X_c$ be a smooth fiber of $f$. 
Then $X_c$ is either a  primary Hopf surface, an Inoue surface, a torus or a primary Kodaira surface with torsion free first homology group. 

\end{lemma} 


 \begin{proof} Note first that $K_{X_c} $ is topologically trivial, since $K_X$ is topologically trivial, due to $b_2(X) = 0$. 
Then we use the tangent sequence 
 $$ 0 \to T_{X_c} \to T_X \vert X_c \to N_{X_c/X} \simeq \sO_{X_c} \to 0 $$
 and observe that $c_2(X) = 0$, since $b_4(X) = 0$. Thus $c_2(X_c) = 0$. 
Since the (sufficiently) general fiber of an algebraic reduction has non-positive Kodaira dimension, \cite[Thm. 12.1]{Ue75},
so does every smooth fiber (see e.g. \cite[VI.8.1]{BHPV04}.
Then we conclude by surface classification
 and using the torsion freeness of $H_1(X_c,\mathbb Z) $, Lemma \ref{lemirrdcomp}.


 \end{proof}


\begin{corollary} 
\label{lem:irrfibers} 
All fibers
 of $f$ are irreducible unless the general fiber of $f$ is a torus or a primary Kodaira surface with torsion free first homology. 
 \end{corollary} 
 
 
We fix some notations for the rest of the paper. 

\begin{notation} \label{not} {\rm 
Let $ S \subset X$ be an irreducible reduced surface. In particular, $S$ is Gorenstein. We denote by $\omega_S$ the dualizing sheaf, which is a line bundle. 
Let 
$$ \eta: \tilde S \to S$$ be the normalization of $S$; denote by $N \subset S$ the non-normal locus, equipped with the
complex structure given by the conductor ideal. Let $\tilde N \subset \tilde S$ be the complex-analytic preimage. 
Let
$$ \pi: \hat S \to \tilde S$$
be a minimal desingularization and 
$$ \sigma: \hat S \to S_0$$
be a minimal model. 
For the class of $\omega_S$ we write $K_S$, analogously for $\omega_{\tilde S}$, etc. 

}
\end{notation} 

\begin{lemma} \label{lem:Mori} In the notations of (\ref{not}), we have
$$ \omega_{\tilde S} \simeq \eta^*(\omega_S) \otimes \sI_{\tilde N}$$
and 
$$ \omega_{\hat S} \simeq \pi^*\eta^*(\omega_S) \otimes \pi^*(\sI_{\tilde N}) \otimes \sO_{\hat S}(E) =  \pi^*\eta^*(\omega_S) \otimes \sO_{\hat S}(-\hat N) \otimes \sO_{\hat S}(- \hat E)$$
with an effective divisor $E$ supported on the exceptional locus of $\pi$ and $\hat N$ the strict transform of $\tilde N$ in $\hat S$.
Moreover, there are exact sequences
$$ 0 \to \sO_S \to \eta_*(\sO_{\tilde S}) \to \omega^{-1}_S \otimes \omega_N \to 0$$
and
$$ 0 \to \sO_N \to \eta_*(\sO_{\tilde N}) \to \omega^{-1}_S \otimes \omega_N \to 0$$
\end{lemma} 

\begin{proof} \cite[chap.3, sect.8]{Mo82}. 
\end{proof} 

As an immediate consequence, we note 

\begin{proposition} \label{prop:coho}  Let $S$ be any irreducible reduced compact Gorenstein surface with $\omega_S \equiv 0 $ and $\chi(S,\sO_S) = 0.$ Then 
\begin{enumerate}
\item $\chi(\tilde S,\sO_{\tilde S}) =  \chi(N,\omega^{-1}_S \otimes \omega_N) = - \chi(N,\sO_N)$;
\item $\chi(\tilde N,\sO_{\tilde N}) = 0.$
\end{enumerate} 
\end{proposition} 

\begin{proof} The first equation in (a) follows from Lemma \ref{lem:Mori}. As to the second equation in (a), observe by Serre duality
$$ \chi(N,\omega_S^{-1} \otimes \omega_N) = - \chi(N,{\omega_S}_{\vert N})  = - \chi(N,\sO_N),$$
since $\omega_S \equiv 0$. For the same reasons
$$ \chi(\tilde N,\sO_{\tilde N}) = \chi(N,\sO_N) + \chi(N,\omega_S^{-1} \otimes \omega_N) = 0.$$
\end{proof} 

 
 \begin{proposition} \label{cor:VF}
 Let $X_c$ be a smooth fiber of $f$. Then
 $$ H^0(X_c,{T_X}_{ \vert X_c}) \ne 0.$$

 
 \end{proposition}
 
 \begin{proof}  Consider the exact sequence
 $$ 0 \to H^0(X_c,T_{X_c}) \to H^0(X_c,{T_X}_ {\vert X_c}) {\buildrel {\kappa} \over {\longrightarrow} } H^0(X_c, N_{X_c/X}) \simeq \mathbb C.$$ 
 If $H^0(X_c,T_{X_c}) \ne 0$, the assertion is clear. So it remains to treat the case that $X_c$ has no vector fields. By Lemma \ref{lem:fibers} and \cite{In74}, 
 $X_c$ is an Inoue surface of type $S_M$ or $S_N^{-}$, in which cases $H^1(X_c,T_{X_c}) = 0$. Thus $X_c$ is rigid and $\kappa $ is surjective, so that 
 $H^0(X_c,T_X \vert X_c) \ne 0$ also in these cases. 
 
 \end{proof} 
 
 \begin{corollary} \label{cor:VF2} Let $X_c = \lambda S$ be a fiber with $S$ an irreducible singular surface and $\lambda \geq 1$.
 Then there exists a finite \'etale cover $S' \to S$ such that $H^0(S',T_{S'}) \ne 0$.
  \end{corollary} 
  
  \begin{proof} We consider the tangent sequence
  $$ 0 \to T_S \to {T_X}_{ \vert S}  {\buildrel \kappa \over {\longrightarrow}} N_{S/X}.$$ 
  If $\lambda = 1$, then $N_{S/X} \simeq \sO_S$. However, $\kappa $ is not surjective, since $S$ is singular.
  Hence 
  $$ H^0(S,T_S) = H^0(S,{T_X}_{ \vert S}).$$
  By semicontinuity and Proposition \ref{cor:VF}, $H^0(S,T_X \vert S) \ne 0$ and we conclude. \\
  If $\lambda \geq 2$, arguing in the same way, we obtain a torsion line bundle $\sL$ on $X$ such that 
  $$ H^0(S,T_S \otimes \sL_{\vert S}) \ne 0.$$
  Then we pass to a finite \'etale cover $\tilde S \to S$ to trivialize $\sL_{\vert S}$. 
  
  \end{proof} 
  
  \begin{remark} \label{rem:VF} A vector field $v \in H^0(S,T_S)$ induces canonically a vector field $v_0 \in H^0(S_0,T_{S_0})$. We shortly say that $v_0$ comes from $S$. 
  \end{remark}




\begin{proposition} \label{prop:normal}
Let $X_c = \lambda S$ with $S$ singular, irreducible. Then $S$ is non-normal. 
\end{proposition}  

\begin{proof} Suppose $S$ normal and let $\pi: \hat S \to S$ be a minimal desingularization and $\sigma: \hat S \to S_0$ a minimal model. \\
(a) Suppose that $S$ has only rational singularities, hence $S$ has only rational double points. 
Then $K_{\hat S} \equiv 0$, in particular $\hat S$ is a minimal surface containing $(-2)$-curves. 
By surface classification, $\hat S$ is either a K3 surface, an Enriques surface, of type VII or non-K\"ahler of Kodaira dimension $\kappa (\hat S) = 1$. The first two cases are impossible since
$$ \chi(\hat S,\sO_{\hat S}) = \chi(S,\sO_S) = 0.$$
If $\hat S$ is of type VII, then it must be a Hopf surface or an Inoue surface, since $K_{\hat S} \equiv 0$, but these surfaces do not contain 
$(-2)$-curves.  If $\kappa (\hat S) = 1$, then, since $\hat S$ has a vector field, it does not any rational curve, see e.g. \cite[Satz 1]{GH90}. 



\medskip 
(b) Suppose now that $S$ has at least one irrational singularity. 
Then $\chi(\hat S,\sO_{\hat S}) < \chi(S,\sO_S) = 0, $
hence
$$ h^1(S_0,\sO_{S_0}) =  h^1(\hat S, \sO_{\hat S}) \geq 2.$$ 
Suppose first that $S_0$ is not K\"ahler. By classification, $S_0$ has to be a primary Kodaira
 surface or $\kappa (S_0) = 1$. By Corollary \ref{cor:VF2}, $H^0(S_0,T_{S_0}) \ne 0$ (up to finite \'etale cover).
Choose a nonzero vector field $v_0$ coming from $S$; then $v_0$ does not have zeroes by classification, \cite[Satz 1]{GH90}; note that in case $\kappa (S_0) = 1$, $S_0$ is an elliptic bundle over a curve of 
genus at least $2$. Hence we must have $\hat S = S_0$.
But then $\hat S$ does not contain contractible curves, so that $S$ is smooth, a contradiction. \\
Thus $S_0$ is K\"ahler. Since $K_{\hat S} = \pi^*(K_S) - E$ with $E$ a non-zero effective divisor,  $\kappa (S_0) = -\infty$ 
and $S_0$ is a ruled surface over a curve $B$ of genus $g = g(B) \geq 2$. 
Since $S$ has an irrational singularity, $\pi$ must contract an irrational curve whose normalization necessarily has genus at least $g$.
Thus $$h^0(S,R^1\pi_*(\sO_{\hat S})) \geq g$$
and therefore
$$1-g = \chi(\hat S,\sO_{\hat S}) \leq \chi(S,\sO_S) - g = - g,$$
which is absurd. 
\end{proof}



\section{Kodaira surfaces and tori} 

In this section we consider the case that the general fiber of $f$ is a Kodaira surface or a torus. 
We rule out the case of Kodaira fibers and show in the torus case that for general line bundles $\sL$ on $X$, the restriction $\sL_{\vert X_c}$ to any fiber is never torsion. 

 
\begin{proposition} \label{prop:localfreeness}  Assume that the general fiber of $f$ is a Kodaira surface or a torus.  Then
$R^jf_*(\sO_X)$ is locally free for all $j$, in fact, $h^j(X_c,\sO_{X_c})$ is independent on $c \in C$.
\end{proposition}

\begin{proof}  It suffices to show that $h^2(X_c,\sO_{X_c})$ is independent of $c$. Indeed, since $h^0(X_c,\sO_{X_c}) = 1 $ for all $c$ and since
$\chi(X_c,\sO_{X_c})$ is constant, $h^1(X_c,\sO_{X_c})$ does not depend on $c $ as well, and the assertions follow by Grauert's theorem. 
By Serre duality, 
$$H^2(X_c,\sO_{X_c}) = H^0(X_c,\omega_{X_c}) = H^0(X_c,\omega_X \vert X_c).$$
Setting $\sL = f_*(\omega_X)$, a locally free sheaf of rank one, we obtain
$$ \omega_X = f^*(\sL) \otimes \sO_X(\sum_i (m_i -1) F_i),$$
where $F_i$ are the non-reduced fiber components. In particular, ${\omega_X}_{ \vert X_c} = \sO_{X_c} $ for all reduced fibers $X_c$ and therefore 
$$ h^0(X_c,\omega_{X_c}) = 1 $$
for all those $c$. 
So let $X_c$ be a non-reduced fiber and set $Y = {\rm red}(X_c)$. 
We consider the complex subspace 
$$Z = \sum (m_i-1)  F_i $$ of $X_c$ and have an induced exact sequence
$$ 0 \to \sI_{Z/X_c} \otimes {\omega_X}_{\vert X_c} \simeq \sO_{X_c} \to {\omega_X}_{\vert X_c} \to {\omega_X}_{\vert Z} \to 0.$$ 
Applying $f_*$ and observing that 
$$ f_*(\sO_{X_c}) = f_*({\omega_X}_{\vert X_c} ) = \sO_{\{c\}},$$
shows that the restriction map
$$ H^0(X_c , {\omega_X}_{\vert X_c}) \to H^0(Z, {\omega_X}_{\vert Z}) $$
vanishes. 
Since 
$$ \dim  H^0(X_c , {\omega_X}_{\vert X_c}) = \dim H^0(X_c,\sO_{X_c}) = 1,$$
we conclude  $ h^0(X_c,\omega_{X_c}) = 1$.





%Let $m$ be the smallest integer such that the $m$-th infinitesimal neighborhood $Y_m$ of $Y $ in $X_c$ 
%is $X_c$. 
%Then $$ H^0(Y, {\omega_{X_c}}_{\vert Y} \otimes N^{*\mu}_{Y/X_c})  = H^0(Y, {\omega_{X}}_{\vert Y} \otimes N^{*\mu}_{Y/X_c})  = 0$$
%for $0 \leq \mu \leq  m-1$ and 
%$$ h^0(Y, {\omega_{X_c}}_{\vert Y} \otimes N^{*m}_{Y/X_c})  = 1,$$
%so that $ h^0(X_c,\omega_{X_c}) \leq 1,$
%hence  $ h^0(X_c,\omega_{X_c}) = 1$ by semicontinuity. 

\end{proof} 

\begin{corollary} \label{cor:restrict}  Let $F$ be a general smooth fiber of $f$. 
Then the restriction map
$$ H^1(X,\sO_X) \to H^1(F,\sO_F) $$
is surjective. 
\end{corollary}

\begin{proof} This is Theorem 3.1 in \cite{CDP98}; the proof works since we now know that $R^1f_*(\sO_X)$ is locally free. 
\end{proof} 

As a consequence, we obtain 

\begin{corollary} \label{cor:Kodaira}

The general fiber of $f$ cannot be a Kodaira surface.
\end{corollary}

\begin{proof} This is  Proposition 3.6 in \cite{CDP98}. In the proof  of Proposition 3.6, Theorem 3.1 is used which is now established by Corollary \ref{cor:restrict}. 
Notice that in Step 2 of the proof of Proposition 3.6 in \cite{CDP98}, the local freeness of $R^jf_*(\tilde \sL)$ is used only generically. 
\end{proof}  

\begin{remark} The same arguments also rule out Hopf surfaces of algebraic dimension one. 
\end{remark}

From now - for the remainder of this section - we assume that the general of $f$ is a torus. 


\begin{proposition} \label{prop:triv}
$R^1f_*(\sO_X) ) \simeq  \sO_C(b_1)  \oplus \sO_C(b_2)$ with $b_j \geq 0$. 
\end{proposition} 

\begin{proof}  By Proposition \ref{prop:localfreeness}, the sheaf $R^1f_*(\sO_X)$ is locally free of rank two.
Write
$$ R^1f_*(\sO_X) = \sO_C(b_1) \oplus \sO_C(b_2).$$
We observe that  $R^1f_*(\sO_X)$ is generically spanned by
Corollary \ref{cor:restrict}, since 
$$ H^1(X,\sO_X) = H^0(C,R^1f_*(\sO_X)).$$ 
Hence $b_j \geq 0$. 

\end{proof} 

\begin{proposition} \label{prop:torsion1} 
For general  $\sL \in {\rm Pic}(X)$, the restriction $\sL_{\vert X_c}$ is not torsion for all $c \in C$. 
\end{proposition} 
\begin{proof} 
By Proposition \ref{prop:triv}, $h^1(X,\sO_X) \geq 2 $ and the restriction 
$$ H^1(X,\sO_X) \to H^1(X_c,\sO_{X_c}) $$
is surjective for all $c$. Consequently, the kernel of the restriction 
$$ {\rm Pic}(X) = H^1(X,\sO_X) \to {\rm Pic}^{\circ}(X_c) = H^1(X_c,\sO_{X_c}) /H^1(X_c,\mathbb Z) $$
is discrete for all $c$ plus a linear subspace of codimension $2$.  Since $\dim C = 1$, 
it follows that for $\sL \in {\rm Pic}(X)$ general, the restriction $\sL_{ \vert X_c}$ is never trivial and thus also not torsion. 
\end{proof} 





\section{Hopf and Inoue surfaces} 

In this section we assume that the general fiber of $f$ is a Hopf or Inoue surface and show that for general line bundles $\sL$ on $X$, the restriction $\sL_{\vert X_c}$ is never torsion.

\begin{proposition} \label{prop:torsion2}
Assume that the general fiber of $f$ is a Hopf or Inoue surface. Let $\sL \in {\rm Pic}(X)$ be general. Then $\sL_{\vert X_c}$ is not torsion for all $c \in C$, and the restriction map 
${\rm Pic}(X) \to {\rm Pic}(X_c)$ is surjective for any smooth fiber $X_c$. 
\end{proposition} 

\begin{proof} The exact sequence
$$ 0 \to \mathbb Z \to \mathbb C \to \mathbb C^* \to 1 $$
and our assumptions give 
$$ H^1(X,\mathbb C^*) = 0.$$ 
Moreover, $H^2(X,\mathbb C^*)$ is torsion. Consider the canonical morphism
$$\lambda: H^0(C,R^1f_*(\mathbb C^*)) \to H^2(C,\mathbb C^*) \simeq \mathbb C^*.$$
Then by the Leray spectral sequence, $\lambda $ is injective and the cokernel is torsion.
Hence
$$ H^0(C,R^1f_*(\mathbb C^*)) \simeq \mathbb C^*.$$
Choose
$$ 1 \ne u \in H^0(C,R^1f_*(\mathbb C^*)) $$
non-torsion. 
This section defines an inclusion
$$ \iota: \mathbb C^* \to R^1f_*(\mathbb C^*). $$
Let $C_0$ be the smooth locus of $f$ in $C$.
We claim that 
\begin{equation}\label{eq(*)} R^1f_*(\mathbb C^*) \vert C_0 = R^1f_{0*}(\mathbb C^*) \simeq \mathbb C^*. \end{equation} 
Suppose first that Claim (\ref{eq(*)}) holds. Then we conclude as follows. Certainly, $\iota $ is an isomorphism over $C_0$. Thus
we obtain a sequence
$$ 0 \to \mathbb C^* \to R^1f_*(\mathbb C^*) \to Q \to 0$$
where $Q$ is supported on the finite set $C \setminus C_0$. Since 
$H^0(C,R^1f_*(\mathbb C^*)) \simeq \mathbb C^*$ and since $H^1(C,\mathbb C^*) = 0$, it follows $H^0(C,Q) = 0$, hence $Q  = 0$. 
Thus $\iota$ is an isomorphism everywhere and consequently $u$ never takes value one, nor does - by our choice of $u$ - any multiple $u^m$. 
Hence $u$ defines a line bundle $\sL$ such that 
$\sL_{\vert X_c}$ is not torsion for all $c \in C$. 
\vskip .2cm \noindent
It remains to prove Claim (\ref{eq(*)}). 
As before, set $\Delta = C \setminus C_0$, $A = f^{-1}(\Delta)$ and $X_0 = X \setminus A$.  Then, as in the proof of Lemma \ref{lemirrdcomp},
$$ H^4(A,\mathbb C^*) = H^1(X_0,\mathbb C^*) = H^1(C_0,\mathbb C^*) \oplus H^0(C_0,R^1f_{0*}(\mathbb C^*)).$$
Since $H^4(A,\mathbb C^*) \simeq (\mathbb C^*)^s$, it follows
$$ H^0(C_0,R^1f_{0*}(\mathbb C^*)) \simeq \mathbb C^*.$$ Since 
$R^1f_{0*}(\mathbb C^*)$ is locally constant of rank one, the claim follows. 

\end{proof} 

As a consequence we obtain 


\begin{corollary} \label{cor:hopf-inoue} \begin{enumerate} 
\item $R^1f_*(\sO_X) \simeq \sO_C$;
\item $R^2f_*(\sO_X) = 0$;
\item for general $\sL \in {\rm Pic}(X)$ and all $c \in C$, we have $$H^0(X_c, \sL_{\vert X_c}) = H^0(X_c, \sL^*_{\vert X_c}) = 0.$$
%\item ${\omega_X}{\vert X_c}$ is not torsion for all $c \in C$. 
\end{enumerate}
\end{corollary}

\begin{proof} (a) Since $R^1f_*(\sO_X)$ has rank one, we may write
$$ R^1f_*(\sO_X) \simeq \sO_C(a) \oplus {\rm torsion}.$$
By Proposition \ref{prop:torsion2}, $a = 0$. So it remains to show that $R^1f_*(\sO_X)$ is torsion free. If not, there exists a line bundle $\sM$,
such that $\sM_{\vert X_{c_0}}$ is not torsion for some $c_0$ but $\sM_{\vert X_c} \simeq \sO_{X_c}$ for $c \ne c_0$. 
Write $ S = {\rm red}(X_{c_0})$. 
Then
$$\sM = f^*f_*(\sM) \otimes \sO_X(D),$$
with an effective divisor $D$ supported on $S$. Since $S$ is irreducible, $D = mS$ and therefore $\sO_X(D)_{\vert X_c}$ is a torsion line bundle,
contradiction. \\
(b) By (a), $h^1(X,\sO_X) = 1$. Since the general fiber of $f$ having negative Kodaira dimension, we have
 $$H^3(X,\sO_X) = H^0(X,\omega_X) = 0.$$
Thus we conclude from $\chi(X,\sO_X) = 0 $, that
$$H^2(X,\sO_X) = 0.$$
Hence, by the Leray spectral sequence, $R^2f_*(\sO_X)$ must be torsion free, therefore
$$ R^2f_*(\sO_X) = 0.$$ 
(c) As a consequence of (b), $R^2f_*(\sL) = 0$ for general $\sL$, hence 
$$ H^2(X_c,\sL_{X_c}) = 0$$
for all $c$. Thus 
$$ H^0(X_c,\sL_{X_c}) = 0$$
for general $\sL$ and all $c$ as well. 
In summary, we may say that 
$$ H^0(X_c,\sL_{\vert X_c}) = H^0(X_c,\sL^*_{\vert X_c})$$
for general $\sL$ and all $c$. \\
%(d) As in the proof of Proposition \ref{prop:torsion2} for $\mathbb Z$ instead of $\mathbb C^*$,  we deduce that 
%$$ R^1f_*(\mathbb Z) \simeq \mathbb Z. $$ 
%The Leray spectral sequence yields
%$$ \mathbb C \simeq H^1(X,\sO_X^*) \simeq H^1(C,\sO_{C}^*) \oplus H^0(C,R^1f_*(\sO_X^*),$$
%hence
%$$  H^0(C,R^1f_*(\sO_X^*) \simeq \mathbb C^*,$$
%and therefore 
%$$ R^1f_*(\sO_X^*) \simeq \sO_C^*.$$
%Redundant: 
%Applying direct images to the exponential sequence on $X$ yields
%$$ 0 \to R^1f_*(\mathbb Z) \to R^1f_*(\sO_X) { \buildrel \kappa \over {\longrightarrow}} R^1f_*(\sO_X^*) \to R^2f_*(\mathbb Z) \to 0,$$ 
%which by our previous considerations reduces to
%$$ 0 \to \mathbb Z \to \sO_C \to \sO_C^* \to R^2f_*(\mathbb Z) \to 0,$$
%and hence the torsion sheaf $R^2f_*(\mathbb Z) = 0$.
%Now $\omega_X^{m} $ defines sections $s_m \in H^0(C,R^1f_*(\sO_X))$ and $t_m \in H^0(C,R^1f_*(\sO_X^*))$ such that $t_m = \kappa (s_m)$. 
%Notice that for $c \in C_0$, the smooth locus of $f$, the bundle $\omega^m_X_{\vert X_c) = \omega^m_{\vert X_c})$ is never trivial. 
%Thus the constant section $s$ does not take integer values on $C_0$, hence on $C$. Thus $t$ never takes value $1$, and our claim follows. \\
Now $\omega_X^{m} $ defines a section $t_m \in H^0(C,R^1f_*(\sO_X^*))$. Notice that for $c \in C_0$, the smooth locus of $f$, the bundle $\omega^m_{X_{\vert X_c}} = \omega^m_{\vert X_c})$ is never trivial and thus
$t_m$ does not take value $1$ on $C_0$. 
Since $R^1f_*(\sO_X^*) \simeq \sO_C^*$, the section never takes
value $1$, hence our claim follows. 
\end{proof} 

We will further need the following basic statement on Hopf and Inoue surfaces. 

\begin{proposition} 

\label{prop:easy}

Let $S$ be a primary Hopf surface. 
Assume that  $$ H^0(S,\Omega^1_S \otimes \sL) \ne 0 $$
for some line bundle $\sL$ on $S$. Then 
$$ H^0(S,\sL) \ne 0.$$
\end{proposition} 

\begin{proof} Choose a vector field $v$ on $S$ and let $C$ be the zero locus of $v$ which is purely one-dimensional. We obtain an
exact sequence
$$ 0 \to \sO_S(C) \to T_S \to \sO_S(-C) \otimes \omega^{-1}_S \to 0.$$
Dualizing, 
$$0 \to \sO_S(C) \otimes \omega_S  \to \Omega^1_S \otimes \sO_S(-C) \to 0.$$
Hence 
$$ H^0(S,\sO_S(C) \otimes \omega_S \otimes \sL) \ne 0$$
or 
$$H^0(S,\sO_S(-C) \otimes \sL) \ne 0,$$
In the latter case, the claim is clear. In the first case we observe that $$H^0(S,\omega^{-1}_S \otimes \sO_S(-C)) \ne 0.$$
Indeed, there exists another vector field $v'$, and 
$v \wedge v'$ is a section of $\omega^{-1}_S$ vanishing on $C$. 

\end{proof} 

\begin{proposition} \label{prop:easy1}
Let $S$ be an Inoue surface. Then there is a unique line bundle $\sL$ such that
$$ H^0(S,\Omega^1_S \otimes \sL) \ne 0.$$
Moreover, one of the following statements holds. 
\begin{enumerate}
\item either $H^0(S,T_S) \ne 0$ and $\sL = \omega_S^{-1}$
\item  or $H^0(S,T_S) = 0$ and $\sL^{\otimes 2} \simeq \omega^{-1}_S$. 
\end{enumerate} 

\end{proposition} 


\begin{proof}  The existence of $\sL$ is classical, \cite{In74}. \\
If $S$ has a non-zero vector field $v$, necessarily without zeroes, then $v$ induces an exact sequence
$$ 0 \to \sO_S \to T_S \to \omega_S^* \to 0, $$
and the claim is immediate, since $S$ has no curves and since $H^0(S,\Omega^1_S) = 0$. 
If $H^0(S,T_S) = 0$, consider the exact sequence
$$ 0 \to \sL^* \to \Omega^1_S \to \sL \otimes \omega_S \to 0.$$ Since the sequence does not split, 
$$ H^1(S,\omega^{-1}_S \otimes  \sL^{\otimes -2}) \ne 0.$$
Hence either $\omega^{-1}_S \simeq \sO_{\otimes 2}$ or $ \omega_S \simeq \sL$, \cite[Lemma 1]{In74}. 
The second case however cannot happen, since $$H^0(S,\Omega^1_S \otimes \otimes \omega_S \simeq H^2(S,T_S) \ne 0,$$
\cite[Prop.2]{In74}. 

\end{proof} 











\section{Proof of Theorem \ref{MT2} }

As already said in the introduction, it suffices to prove Proposition \ref{prop:CDP}. 
Thus we need to show that 
$$ H^2(X_c,{T_X}_{\vert X_c} \otimes \sL_{\vert X_c}) =  0$$
for all $c \in C$. 
By Serre duality, this comes down to show
that 
$$ H^0(X_c, {\Omega^1_X}_{\vert X_c} \otimes \sL_{\vert X_c}) = 0$$
for some $\sL \in {\rm Pic}(X) $ and for all $c \in C$. 

We first consider irreducible fibers. Let 
$$ S = {\rm red}X_c.$$
Using the (co-)tangent sheaf sequence
$$ 0 \to N^*_S \to {\Omega^1_X}_{\vert  S} \to  \Omega^1_{S} \to 0 $$
it is immediate that it suffices to show - provided $\sL_{\vert S}$ is not torsion -  the following statement
\begin{equation} \label{eq:main1}  H^0(S,\tilde \Omega^1_S \otimes \sL_{\vert S}) = 0, \end{equation} 
where $$ \tilde \Omega^1_S = \Omega^1_S/{\rm torsion}.$$


We first treat smooth fibers $S = X_c$.

\begin{proposition}  Equations (\ref{eq:main1})  holds for smooth fibers $S$ (independent on the structure of the general fiber), i.e., for  $\sL \in {\rm Pic}(X) $ general 
$$  H^0(S, \Omega^1_S \otimes \sL_{\vert S}) =  0 $$
simultaneously for all smooth fibers $S$. 
\end{proposition} 

\begin{proof}
%Since ${\rm Pic}(X) \to {\rm Pic}^{\circ}(S)$ is surjective, we only to consider line bundles $\sL$ on $S$. 
%Set $\sM = \sL_{\vert S}$, hence $\sM \in {\rm Pic}^{\circ}(S)$ is not torsion and $H^0(S,\sM) = H^0(S,\sM^*) =  0$. \\
(a) First, if $S$ is a torus, then 
$$ H^0(S,T_S \otimes \sA)  = H^0(S,\Omega^1_S \otimes \sA) = 0$$
for all non-trivial $\sA$, hence we may take any $\sL$ such that $\sL_{\vert X_c}$ is never trivial, Proposition \ref{prop:torsion1}. \\
%(b) If $S$ is a Hopf surface with $a(S) = 1$, then there is an elliptic fiber bundle $p: S \to B \simeq \mathbb P_1$,
%leading to an exact sequence
%$$ 0 \to p^*(\sO_B(-2)) \to \Omega^1_S \to \omega_{S/B} \simeq \sO_S \to 0.$$ 
%If $\sM \ne p^*(\sO_C(a)$ with $a \geq 0$, 
%then clearly there cannot be an non-zero map $\sM^* \to \Omega^1_S$. 
%The $T_S$-case is similar, dualizing the cotangent sequence. \\
%(c) Suppose next that $S$ is a (primary) Hopf surface with $a(S) = 0$.  Consider an inclusion 
%$\sL^* \to T_S$ and a vector field $v \in H^0(T_S$ with zero locus $C$. This leads to a sequence
%$$ 0 \to \sO_S \to T_S \to \sI_C \otimes \omega^*_S \to 0.$$
%The induced maps $\sM^* \to \sI_C \otimes \omega^*_S$ vanishes if $\sM \ne \sO_S$ (by the structure of divisors on Hopf surfaces). 
%Thus we obtain an inclusion $\sM^* \to O_S$ so that $\sM$ is of the form $\sO_S(D)$ with $D$ an effective divisor, which is forbidden.  \\
%(d) Finally, let $S$ be an Inoue surface. Then there exists a topologically trivial line bundle $\sA$ such that $H^0(S,T_S \otimes \sA)$. 
%Note that $S$ does not contain any curve. Hence, if $\sM^* \to T_S$ is non-zero, then either $\sM^* \simeq \omega^*_S \otimes \sA^*$
%or $\sM^* \simeq \sA^*$. 
(b) If $S$ is a Hopf surface, then by Corollary  \ref{cor:hopf-inoue}, $H^0(S,\sL_{\vert S}) = 0$ for general $\sL$, hence we conclude by
Propositions \ref{prop:torsion2} and \ref{prop:easy}. \\
(c) If $S$ is an Inoue surface with a vector field, then for $\sL$ general, also $\sL^* \otimes \omega_X$ is general, hence 
$$ H^0(S,\sL^*_{\vert S} \otimes (\omega_X )_{\vert S}) = H^0(S,\sL^* \otimes \omega_S)) = 0,$$
hence we conclude by Propositions \ref{prop:torsion2} and \ref{prop:easy1}. \\
(d) Finally, assume that $S$ is an Inoue surface without vector field. Then we argue as in (c), observing that $(\sL^*)^{\otimes 2} \otimes \omega_X$ is general for
general $\sL$. 
\end{proof} 


\begin{remark} \label{rem:torsion} 
Since the conormal bundle of a multiple fiber is torsion, the arguments also apply to fibers $X_c = \lambda S $ with $\lambda \geq 2$ and 
$S$ smooth. 
\end{remark} 

\begin{proposition} \label{prop:1}  Equation (\ref{eq:main1}) holds for singular reduced fibers.
\end{proposition} 

\begin{proof}  Recall the notations \ref{not}.  By Lemma \ref{lem:Mori} and Proposition \ref{prop:normal}, $\kappa (S_0 ) = - \infty$, the surface $S$ is non-normal and 
$$ H^2(\tilde S,\sO_{\tilde S}) = H^0(\tilde S,\omega_{\tilde S}) = 0.$$ 
Arguing by contradiction, there exists a one-dimensional family $\sL_t $ of line bundles on $X$ such that $$H^0(S, \tilde \Omega^1_S \otimes {\sL_t}_{\vert S}) \ne 0.$$
Passing to a desingularization and then to a minimal model $S_0$ as in Notation \ref{not}, 
there are numerically trivial line bundles $\sM_t$ on $S_0$ such that 
$$\sM_t = \sigma_* \pi^*\eta^*(\sL_t)^{**} $$
with a one-dimensional family of sections in 
$$ H^0(S_0,\Omega^1_{S_0} \otimes \sM_t).$$
Thus 
\begin{equation} \label{eq:non}  H^0(S_0,\Omega^1_{S_0} \otimes \sM_t) \ne 0. \end{equation} 
Observe that all line bundles  $\sM_t$ might be trivial. 
\vskip .2cm \noindent {\it Step 1.} 
Suppose first that $S_0$ is K\"ahler. Then by (\ref{eq:non}),  
$S_0$ must be ruled over a curve of genus at least two. 

{\it Claim.  $\tilde S$ has rational singularities, only.} 


{\it Proof of the Claim.} Assume to the contrary that  $\tilde S$ has an irrational singularity. We claim that $H^1(\tilde S,\sO_{\tilde S}) = 0$. In fact, $\pi$ must contract a curve  $B_0$
projecting onto $B$. Thus $h^1(B_0,\sO_{B_0}) \geq g$ and therefore $h^0(\tilde S, R^1\pi_*(\sO_{\hat S})) \geq g$. Since $H^2(\tilde S,\omega_{\tilde S}) = 0$, the Leray spectral sequence
yields $H^1(\tilde S,\sO_{\tilde S}) = 0$ (and $h^0(\tilde S, R^1\pi_*(\sO_{\hat S})) = g$).

Thus all line bundles $\eta^*(\sL_t) $ are trivial
and we obtain a one-dimensional family $\tilde \omega_t$ of holomorphic one-forms on $\tilde S$. Moreover there exists a one-dimensional family $\omega_t$  of one-forms
on $B$ such that
$$ \sigma^* p^*(\omega_t) = \pi^*(\tilde \omega_t),$$
where $p: S_0 \to B$ is the ruling. 
Since $p(\sigma(B_0)) = B$, we have $\iota_{B_0}^*\sigma^* p^*(\omega_t)\ne 0$.
 On the other hand, since $\pi$ contracts $B_0$, it follows that $\iota_{B_0}^*(\pi^*(\omega_t)) = 0$, a contradiction.  This proves the {\it Claim} and thus $\tilde S$ has rational singularities, only. 


In this case the morphism $p_0: S_0 \to B$ induced a morphism $\tilde p: \tilde S \to B$. 
In the language of divisors and using the notations of (\ref{not}) and (\ref{lem:Mori}) we have
$$ -K_{\hat S} \equiv \hat N + \hat E,$$
where $\hat N $ is the strict transform of $\tilde N$ in $\hat S$. 
Set 
$$ N_0 = \sigma_*(\hat N).$$
We are now using the theory of ruled surfaces as in \cite[section V.2]{Ha77}, taking over also the notations from \cite{Ha77}. 
In particular we have the invariant $e$ and a section $C_0$ with minimal self-intersection $C_0^2 = -e$. 
Moreover,
$$ -K_{S_0} \equiv 2C_0+ (e+2-2g)F,$$
where $F$ is a fiber of $p_0$ and $g = g(B) $ the genus of  $B$. 
Since $\tilde S$ has rational singularities, $\pi$ cannot contract any curve projecting onto $B$. Hence we must have
$$ N \equiv 2C_0 + aF$$ 
with $a \leq e+2-2g$.
Taking into account the numerical description of irreducible curves in $S_0$, as given in  \cite[section V.2]{Ha77}, it follows immediately that 
$e > 0 $ and that 
$$ N = 2C_0 +R$$
with an effective divisor $R \sim aF$ (note that the curve $C_0$ is the unique contractible curve in $S_0$). 
Consequently, $\tilde N$ has a unique component, say $\tilde N_1$, projecting onto $B$, and this component has multiplicity two. 
The map $\tilde p: \tilde S \to B$ induces a holomorphic map $p: S \to B'$ and a commutative diagram 
$$
\xymatrix{ 
\tilde S \ar[d]_{\tilde p} \ar[r]^{\eta} & S \ar[d]^{p}\\
B\ar[r]_{\tau} & B'.}
$$



The general fiber $S_b$ of $p$ is a reduced Gorenstein curve with 
$$ \omega_{S_b} \equiv 0$$ 
whose normalization of $S_b$ is a disjoint union of  smooth rational curves. Thus, if $S_b$ is irreducible, then $S_b$ is a rational curve with one
node or cusp, and if $S_b$ is reducible, it is a cycle of smooth rational curves. In case $S_b$ has a node or is a cycle, the normalization map 
$\eta$ is generically $2:1$ along $\tilde N_1$. In these cases however, $\tilde N_1$ would be reduced, see \cite{KW88}, a contradiction.  In the remaining case, 
$\eta$ has degree one along $\tilde N_1$, hence $\tau $ has degree one, too. Unless $\tau $ is an isomorphism and $g(B) = 2$, we have 
$h^1(B',\sO_{B'}) \geq 3$, hence $h^1(S,\sO_S) \geq 3$. Since $\chi(S,\sO_S) = 0$, we conclude
$$ h^0(S,\omega_S) = h^2(S,\sO_S) \geq 2.$$ 
Since $S$ is Moishezon and $\omega_S \equiv 0$, this is impossible. 
Alternatively, apply Proposition \ref{prop:triv}  or Corollary \ref{cor:hopf-inoue}, respectively.

Hence $\tau $ is biholomorphic, i.e., $p$ maps to the smooth curve $B$ of genus two and $h^1(S,\sO_S) = h^1(B,\sO_B) = 2$. 
Moreover, $h^0(S,\omega_S) = 1$, and therefore $\omega_S \simeq \sO_S$. 
The map $p$ being flat, $R^1p_*(\sO_S) $ is locally free of rank one, and by relative duality, 
$$ R^1p_*(\sO_S) \simeq p_*(\omega_{S/B})^* = \omega_B,$$
hence $$H^0(B,R^1p_*(\sO_S)) \ne 0.$$
But then $h^1(S,\sO_S) > h^1(B,\sO_B)$, a contradiction. This shows that $g(B) \geq 2$ is impossible and concludes the proof in the K\"ahler case. 
 






\vskip .2cm \noindent  {\it Step 2.} 
We thus are reduced to the case that $S_0$ is not K\"ahler. \\
If $S_0$ is of type VII, then $H^0(S_0,\Omega^1_{S_0}) = 0,$ hence $H^0(S_0,\Omega^1_{S_0} \otimes \sM) = 0$ for $\sM$ general, contradicting (\ref{eq:non}).\\
The same argument applies to a secondary Kodaira surface $S_0$. If $S_0$ is a primary Kodaira surface, then the cotangent sequence reads 
$$ 0 \to \sO_{S_0} \to \Omega^1_{S_0} \to \sO_{S_0} \to 0,$$
which immediately gives a contradiction by tensorizing with $\sM_t$. \\
It remains to exclude the case $\kappa(S_0) = 1$. Since $H^0(S_0,T_{S_0}) \ne 0$, (\ref{cor:VF2}) and (\ref{rem:VF}),  the Iitaka fibration $h_0: S_0 \to B$ is an elliptic bundle over a curve of genus $g(B) \geq 2$, \cite[Satz 1]{GH90} and, as already noticed,  
the induced vector field $v_0$ has no zeroes. Hence $\tilde S = \hat S = S_0$. 
Since $\omega_{\tilde S} = \sI_{\tilde N} \otimes \eta^*(\omega_S)$, 
we have
$$ h^2(S,\sO_S) \geq  h^2(\tilde S, \sO_{\tilde S}),$$
hence $h^2(S,\sO_S) \geq 2$, contradicting Proposition \ref{prop:triv}  or Corollary \ref{cor:hopf-inoue}, respectively.

\end{proof} 

\begin{remark}  If the fiber $X_c = \lambda S$ with $S$ an irreducible reduced singular surface and $\lambda \geq 2$, we argue in the same way, passing to a finite \'etale cover. 
\end{remark} 


Finally, we have to treat reducible fibers:


\begin{proposition}  Equation (\ref{eq:main1}) holds for reducible fibers.
\end{proposition} 

\begin{proof} Let $$F = \sum a_iS_i$$ be a reducible fiber.
Arguing by contradiction, there is a one-dimensional family $\sL_t$ of line bundles on $X$ such that 
$$ H^0(F,\Omega^1_X \vert F \otimes (\sL_t)_{\vert F}) \ne 0.$$ 
Hence there exists a number $i_0$ such that 
$$ H^0(S_{i_0}, \tilde \Omega^1_X \vert S_{i_0} \otimes ( \sL_t)_{\vert S_{i_0}}) \ne 0 $$
for all $t$, and therefore 
$$ H^0(S_{i_0}, \tilde \Omega^1_{S_{i_0}} \otimes ( \sL_t)_{\vert S_{i_0}}) \ne 0 $$
Now we argue as in Proposition \ref{prop:1} to obtain a contradiction. One might also use the line 
bundle $\sO_X(-kS_{i_1})$ for $k \gg 0$, where the surface $S_{i_1}$ meets in $S_{i_0}$ in a curve. 


\end{proof} 








































 




%\bibliographystyle{alpha}



%\bibliography{biblio}



\end{document}
 












%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%